Natural Science

## Pubertal Onset and Maintenance in FGFR3 Global Knockout Mice

## Amanda Brown

The following is an excerpt from a longer piece. For full text, please visit https://scholar.colorado.edu/concern/undergraduate\_honors\_theses/w9505121v

#### Lay Summary:

Human reproduction relies on a cascade of hormones beginning in a part of the brain called the hypothalamus. The initial hormone, called gonadotropin releasing hormone (GnRH), initiates the hormonal cascade and allows for reproductive potential. The initial protein released, GnRH, is released from GnRH neurons in the hypothalamus. During fetal development, the production of GnRH neurons relies on certain receptors called fibroblast growth factors to direct the correct placement of GnRH neurons in the brain. If migration of the GnRH neurons fails, one cannot go through puberty and would be incapable of reproducing. One receptor that is not well studied for its role in GnRH neuron development and migration is fibroblast growth factor receptor 3 (FGFR3). This study aims to examine the role of FGFR3 in gonadal maturation and function by characterizing mice with a disruption in the FGFR3 gene, meaning that the FGFR3 receptors do not function well or at all in these transgenic mice. The onset of puberty was assessed by sex hormone-mediated events such as vaginal opening (VO) in females and balanopreputial separation (BPS) in males. Additionally, the reproductive potential of the adult mice was also assessed using viable sperm counts in males and reproductive hormone cyclicity in females. FGFR3 deficient females exhibited delayed vaginal opening and abnormal cyclicity. On

the contrary, FGFR3 deficient male mice exhibited normal timing of BPS and normal sperm concentration compared to traditional mice. These results indicated that females were more adversely affected by FGFR3 deficiency than males. These results provide initial evidence that FGFR3 deficiency can disrupt female reproduction. Further, inactivating FGFR3 mutations may contribute to human reproductive disorders such as hypogonadotropic hypogonadism.

#### Abstract:

Vertebrate reproduction is driven by hormones from the hypothalamic-pituitary-gonadal axis. Previous studies showed that neurons from the hypothalamus secreting gonadotropin-releasing hormone (GnRH) require fibroblast growth factors (FGF) to develop and mature, but the specific role of fibroblast growth factor receptor 3 (FGFR3) in overall reproduction is unknown. This study aims to examine the role of FGFR3 in gonadal maturation and function by characterizing mice with a global inactivating mutation in the FGFR3 gene. Pubertal onset was assessed by sex steroid-mediated events such as vaginal opening (VO) in females and balanopreputial separation (BPS) in males. Adult mice were also examined for estrous cyclicity in females and motile sperm concentrations in males. Female FGFR3 heterozygous global knockout mice ( $FGFR_3^{+/-}$ ) exhibited delayed VO and

abnormal estrous cyclicity with reduced time spent in diestrus. On the contrary, male  $FGFR_3^{+/-}$  mice exhibited normal timing of BPS and motile sperm concentration compared to WT mice. These results indicated that females were more adversely affected by FGFR<sub>3</sub> deficiency than males. It is currently unclear if these female-specific adverse effects are exerted at the level of the hypothalamus, pituitary, or gonad. These results provide initial evidence that FGFR<sub>3</sub> deficiency can disrupt female reproduction. Further, inactivating *FGFR*<sub>3</sub> mutations may contribute to human reproductive disorders such as hypogonadotropic hypogonadism.

#### Introduction:

Vertebrate reproduction requires proper development and function of the endocrine organs within the hypothalamic-pituitary-gonadal (HPG) axis. The cascade of hormones produced by this axis begins with the pulsatile release of gonadotropin-releasing hormone (GnRH) from the neuroendocrine hypothalamus. GnRH travels through the portal system where it acts on gonadotropes of the anterior pituitary and induces the release of the gonadotropins, luteinizing hormone (LH) and follicle stimulating hormone (FSH). Finally, the gonadotropins travel through the general circulation and stimulate the gonads (ovaries and testes) to increase gametogenesis and the production of sex steroids. Effective communication within the HPG axis is essential to ensure adequate pubertal onset and fecundity.

Because GnRH is the most upstream activator of the HPG axis, defects in the GnRH system and HPG axis present a severe obstacle to fertility and reproduction. Therefore, it is of utmost importance to identify the factors needed for the development and maintenance of the GnRH system. There is growing evidence suggesting that the formation of the GnRH system is largely dependent on a class of signaling molecules, fibroblast growth factors (FGFs), and their receptors, fibroblast growth factor receptors (FGFRs), which collectively mediate the genesis, differentiation and migration of GnRH neurons (Chung et al. 2008). The FGF signaling family consists of 22 FGF ligands and 4 tyrosine kinase FGFRs. When an FGF binds the extracellular domain of the FGFR, the receptors dimerize, resulting in the cross phosphorylation of the intracellular domain. The cross phosphorylation induces a phosphorylation cascade, resulting in a variety of cellular effects.

FGF signaling molecules and FGFRs are present in the birthplace of GnRH neurons, the olfactory placode, at the time of neuron fate specification (Tsai et al. 2011). GnRH neurons emerge in the olfactory placode and migrate along the olfactory axons through the cribriform plate, where they reach their final destination, the preoptic area (POA) (Wierman et al. 2011). GnRH neuronal cell bodies remain in the POA while their axons extend to the median eminence. GnRH axon terminals release GnRH peptide into the hypophyseal portal system in a pulsatile fashion to induce the onset and maintenance of reproduction.

Although a critical relationship exists between the ligand FGF8 and receptor FGFR1 to allow proper GnRH neuron development, FGFR<sub>3</sub>, another FGF receptor present on GnRH neurons, has not been studied extensively (Chung et al. 2008). In humans, loss-of-function mutations in FGFR1 are causally linked to hypogonadotropic hypogonadism (HH), a condition characterized by insufficient gonadotropins and therefore incomplete/absent puberty and infertility (Fraietta et al. 2013). However, a correlation between FGFR<sub>3</sub> mutations and HH has not been established in humans. There has been only one clinical report on a loss-of-function mutation in *FGFR*<sub>3</sub>, and no reproductive deficits were noted in the two male subjects studied (Makrythanasis et al. 2014). The correlation between FGFR3 and reproduction in rodents was similarly incomplete. One study suggested that FGFR3 did not affect the development of GnRH

neurons but may be needed to maintain postnatal GnRH neurons, but it did not study the reproductive consequence of *FGFR3* deficiency (Chung et al. 2010).

The objectives of this study were to examine the timing of pubertal onset and adult reproductive function in *FGFR3* global knockout male and female mice. To assess the timing of pubertal onset, external indicators of puberty were examined: vaginal opening and first estrus in females and balanopreputial separation in males. Following pubertal onset, general reproductive functions including estrous cyclicity in females and motile sperm concentration in males were assessed by vaginal cytology and epididymal squash, respectively. These results should establish, for the first time, a direct correlation between *FGFR3* deficiency and reproduction that may be extrapolated to humans harboring inactivating *FGFR3* mutations.

#### **Materials and Methods:**

Transgenic animals

C<sub>57</sub>BL/6 mice with a global heterozygous deletion in the *FGFR*<sub>3</sub> gene were obtained from the Jackson Laboratories in Bar Harbor, ME and further propagated by breeding. ... Deletion of these components resulted in a truncated nonfunctional FGFR<sub>3</sub>.

[...]

#### Phenotypic pubertal assessment

Male mice were checked for balanopreputial separation (BPS) by gently retracting the prepuce from the glans penis from PN20 until sufficient separation was apparent. ...

Female mice were examined for the age of vaginal opening (VO) by visual examination of the vulva from PN20 until a clear opening is present. ...

#### Vaginal cytology

Once VO in female peripubertal mice (around PN<sub>30</sub>) had occurred, females were examined daily for vaginal cytology to determine their estrous cycle stages.

[...]

Motile sperm concentration measurement by epididymal squash

At PN60, male mice were anaesthetized using isoflurane and rapidly decapitated. A V-shaped abdominal incision was made, and epididymal fat pads were gently pulled to expose the testes and epididymis. The epididymis was isolated from each testis and placed in a petri dish with 2 mL of prewarmed (37 °C) phosphate-buffered saline (PBS). Both epididymides were used for this procedure. The epididymal cauda was isolated from the rest of the epididymis and vas deferens, minced with two scalpel blades and incubated in 2 ml of PBS at 35 °C for 15 minutes with gentle agitation. Following, 500 µl of sperm-containing PBS was mixed with 500 µl of 3% NaCl to immobilize sperm for consistent counting. The solution containing immobilized sperm was then vortexed and a 10-µl aliquot was used for counting by a hemacytometer. Of the 25 grids, four corner grids and one center grid were counted for number of sperm. Sperm that were not fully enclosed in the grid were only counted if they passed over the top or left line of the grid. Average number of sperm contained in the 5 grids was calculated. The average number of sperm was multiplied by the dilution factor, the number of total grids, and the volume of the hemocytometer to result in a final concentration of sperm per mL.

#### Statistical analysis

Homoscedasticity was observed for all data, thus differences between genotypes were determined by Students *t*-test. Differences were considered significant when p<0.05.

## **Results:**

Phenotypic pubertal assessment

First estrus



[...]



**Figure 1.** (A) Age of VO in *FGFR*<sub>3</sub><sup>+/-</sup> females is significantly delayed compared to WT females (N=8-9). (B) Age of BPS is not significantly different between *FGFR*<sub>3</sub><sup>+/-</sup> and WT males (N=4-5). All data represent mean  $\pm$  SE.





## 132 | Honors Journal 2021

#### Days in estrus

[...]



Figure 3. (A) Percent of time in proestrus over 15 days is not significantly different between WT and  $FGFR_3^{+/-}$  mice (N=9, N=5). (B) Percent of time in estrus over 15 days is not significantly different between WT and  $FGFR_3^{+/-}$  mice (N=9, N=5). (C) Percent of time in metestrus over 15 days is not significantly different between WT and  $FGFR_3^{+/-}$  mice (N=9, N=5). (D) Percent of time in diestrus over 15 days is significantly different between WT and  $FGFR_3^{+/-}$  mice (N=9, N=5). (D) Percent of time in diestrus over 15 days is significantly different between WT and  $FGFR_3^{+/-}$  mice (N=9, N=5) (p<0.05). (E) Percent of time in an indistinguishable stage over 15 days is significantly different between WT and  $FGFR_3^{+/-}$  mice (N=9, N=5) (p<0.05). (E) Percent of time in an indistinguishable stage over 15 days is significantly different between WT and  $FGFR_3^{+/-}$  mice (N=9, N=5) (p<0.05). (E) Percent of time in an indistinguishable stage over 15 days is significantly different between WT and  $FGFR_3^{+/-}$  mice (N=9, N=5) (p<0.05). (E) Percent of time in an indistinguishable stage over 15 days is significantly different between WT and  $FGFR_3^{+/-}$  mice (N=9, N=5) (p<0.05). All data represent the mean ± SE.

[...]



**Motile Sperm Concentration** 

**Figure 4.** Motile sperm concentration obtained by epididymal squash of the cauda epididymis was not significantly different between PN60 WT and *FGFR3*<sup>+/-</sup> males (N=3-5 respectively). All data represent mean ± SE.

#### **Discussion:**

Characterization of the reproductive function is fundamental to understanding the effects of *FGFR*<sup>3</sup> deficiency on the HPG axis. Previously studies showed that *FGFR*<sup>3+/-</sup>mice were born with a normal set of GnRH neurons but later exhibited a postnatal loss (PN60) of these neurons (Chung et al. 2010). However, the reproductive consequence of this neuronal loss was not examined. In addition, *FGFR*<sup>3</sup> is also expressed in the pituitary and gonad (Sharma et al. 2019, Kaminskas et al. 2019), suggesting it may directly mediate the functions of these two downstream organs. This study aimed to examine the onset of puberty in young FGFR3-deficient mice as well as the maintenance of reproduction in older adults.

External pubertal markers for both female and male mice were assessed. BPS is an androgen-dependent event that coincides with the maturation of the testes and is an external indicator of puberty (Korenbrot et al. 1977). Our results showed that WT and *FGFR3*<sup>+/-</sup> males attained BPS at the same time (Figure 1B), suggesting their circulating androgen levels were likely to be similar. These results suggest that the postnatal loss of GnRH neurons in *FGFR3*<sup>+/-</sup> males may not have yet occurred at the time of BPS (around PN30).

Another indication of puberty, VO, was determined after weaning (PN20) of female mice. Vaginal opening is initiated by a surge of estrogen occurring around the time of puberty in female mice (Ojeda et al. 1994). Although their male counterparts achieved puberty at a normal age,  $FGFR_{3^{+/-}}$  females exhibited delayed VO compared to WT (Figure 1A). These results suggest a potential lack of adequate estrogen production to induce apoptosis of lower vaginal mucosa (Rodriguez et al. 1997), leading to delayed VO. Decreased estrogen levels could result from decreased GnRH neurons found in FGFR3+/- mice at PN60. Interestingly, FGFR3-<sup>/-</sup> mice (N=3) never achieved VO even 72 days after birth (data not shown). These mice died at various ages with the oldest being PN72, yet never underwent VO at the time of death. Absence of VO in these mice could be due to a complete absence of functional *FGFR*<sub>3</sub> allele, resulting in absent or insufficient GnRH neurons and therefore no pubertal onset. That being said, FGFR3 /-mice suffer significant musculoskeletal deformities and may experience issues unrelated to primary GnRH deficiency, such as reduced feeding or enhanced stress, that could suppress the HPG axis (Rivier et al. 1991).

Evaluating the age of first estrus is particularly

relevant to fertility in females because it indicates not just an estrogen surge sufficient to induce VO, but the beginning of ovarian cyclicity. Despite a delayed VO,  $FGFR_3^{+/-}$  and WT females attained the first estrus at the same time. These results suggest that like humans, puberty in mice consists of many stages, and FGFR<sub>3</sub> deficiency may influence the earlier (VO) but not the later (first estrus) stages.

Over a 15-day period, females were examined for estrous cyclicity, including the number of days spent in each stage. WT and  $FGFR_3^{+/-}$  mice spent the same percentage of time in proestrus, estrus, and metestrus (Figure 3A, 3B, 3C). Interestingly, WT spent 8.88% more time in diestrus compared to FGFR3+/- mice (Figure 3D). Additionally, *FGFR*3<sup>+/-</sup>mice spent 21.11% of their cycle in an indistinguishable phase consisting of little to no cells and severely crenated cells (Figure 3E). Abnormal GnRH numbers could disrupt HPG axis signaling and interfere with normal estrous cyclicity. Lack of gonadotropins and sex steroids due to a GnRH deficiency could produce the indistinguishable estrous phase seen in these mice. Ovarian analyses should be performed to examine the morphology of a follicle at this indistinguishable phase. Additionally, levels of gonadotropins and sex steroids should be examined for a potential discrepancy. Abnormal estrous cyclicity can be seen in  $FGFR_3^{+/-}$  mice and may cause a decrease in fertility as compared to WT mice.

Male mice were evaluated for fertility by examining motile sperm concentration in adulthood (PN6o). Despite the previously shown reduction in GnRH neurons in PN6o  $FGFR_3^{+/-}$ males (Chung et al. 2010), motile sperm concentration at PN6o was not significantly different between WT and  $FGFR_3^{+/-}$  males (Figure 4). Overall, male mice appear to be less affected by  $FGFR_3$  deficiency than female mice. The cause of this sex difference is unknown. Multiple studies have reported that females are, in general, more vulnerable to HPG axis disruption due to a need for robust hormonal changes to drive the female cycle. The precise locus along the HPG axis leading to this sex difference remains to be investigated.

Our results showed that FGFR<sub>3</sub> deficiency did not impact male puberty and reproductive function, but significantly disrupted female pubertal onset and cyclicity. These results are novel and suggest FGFR<sub>3</sub> inactivating mutations in humans may similarly impact certain aspects of women's reproduction. Future studies should examine tissue-specific *FGFR*<sub>3</sub><sup>+/-</sup> knockout mice to ensure reproductive defects are due to a suboptimal HPG axis and not secondarily to defects in musculoskeletal functions (Kubota et al. 2020). Additionally, the precise extra-hypothalamic loci of the HPG axis impacted by *FGFR*<sub>3</sub> deficiency, such as the pituitary and the gonad, should be investigated.

#### Works Cited:

- Byers SL, Wiles MV, Dunn SL, Taft RA (2012) Mouse Estrous Cycle Identification Tool and Images. PLOS ONE 7(4): e35538. https://doi. org/10.1371/journal.pone.0035538
- Caligioni C. S. (2009). Assessing reproductive status/stages in mice. *Current protocols in neuroscience, Appendix 4*, Appendix–4I. https://doi.org/10.1002/0471142301.nsa04is48
- Chung, W. C., Matthews, T. A., Tata, B. K., & Tsai, P. S. (2010). Compound deficiencies in multiple fibroblast growth factor signaling components differentially impact the murine gonadotrophin-releasing hormone system. *Journal of neuroendocrinology*, 22(8), 944–950. doi:10.1111/j.1365-2826.2010.02024.x
- Chung, W. C., Moyle, S. S., & Tsai, P. S. (2008). Fibroblast growth factor 8 signaling through fibroblast growth factor receptor 1 is required for the emergence of gonadotropin-releasing hormone neurons. *Endocrinology*, 149(10), 4997–5003. https://doi.org/10.1210/ en.2007-1634
- Colvin, J. S., Bohne, B. A., Harding, G. W., McEwen, D. G., & Ornitz, D. M. (1996, April). Skeletal overgrowth and deafness in mice lacking fibroblast growth factor receptor 3. Retrieved from https://www.ncbi.nlm.nih.gov/pubmed/8630492

Fraietta, R., Zylberstejn, D. S., & Esteves, S. C. (2013). Hypogonadotropic hypogonadism revisited. Clinics (Sao Paulo, Brazil), 68 Suppl

/(Suppl 1), 81-88. https://doi.org/10.6061/clinics/2013(supo1)09

- Kaminskas, B., Goodman, T., Hagan, A., Bellusci, S., Ornitz, D. M., & Hajihosseini, M. K. (2019). Characterisation of endogenous players in fibroblast growth factor-regulated functions of hypothalamic tanycytes and energy-balance nuclei. *Journal of neuroendocrinology*, 31(8), e12750. https://doi.org/10.1111/jne.12750
- Korenbrot CC, Huhtaniemi IT, Weiner RI. Preputial separation as an external sign of pubertal development in the male rat. Biol Reprod. 1977;17:298–303
- Kubota, T., Adachi, M., Kitaoka, T., Hasegawa, K., Ohata, Y., Fujiwara, M., Michigami, T., Mochizuki, H., & Ozono, K. (2020). Clinical Practice Guidelines for Achondroplasia. *Clinical pediatric endocrinology : case reports and clinical investigations : official journal of the Japanese Society for Pediatric Endocrinology*, 29(1), 25–42. https://doi.org/10.1297/cpe.29.25
- Long JA, Evans HM. The estrous cycle in the rat and its associated phenomena. Memories of University of California. 1922;6:1–148.
- Makrythanasis, P., Temtamy, S., Aglan, M.S., Otaify, G.A., Hamamy, H. and Antonarakis, S.E. (2014), A Novel Homozygous Mutation in FGFR3 Causes Tall Stature, Severe Lateral Tibial Deviation, Scoliosis, Hearing Impairment, Camptodactyly, and Arachnodactyly. Human Mutation, 35: 959-963. doi:10.1002/humu.22597
- Ojeda SR, Urbancki HF. Puberty in the rat. In: Knobil E, Neill JD, editors. Physiology of Reproduction. Raven Press; New York: 1994. pp. 363–409.
- Rivier C, Rivest S. Effect of stress on the activity of the hypothalamic-pituitary-gonadal axis: peripheral and central mechanisms. Biol Reprod. (1991) 45:523–32. 10.1095/biolreprod45.4.523
- Rodriguez, I., Araki, K., Khatib, K., Martinou, J., & Vassalli, P. (1997). Mouse Vaginal Opening Is an Apoptosis-Dependent Process Which Can Be Prevented by the Overexpression of Bcl2. *Developmental Biology*, *184*(1), 115-121. doi:10.1006/dbio.1997.8522
- Sharma, S., Schlatt, S., Van Pelt, A., & Neuhaus, N. (2019). Characterization and population dynamics of germ cells in adult macaque testicular cultures. *PloS one*, 14(6), e0218194. https://doi.org/10.1371/journal.pone.0218194
- Wierman, M. E., Kiseljak-Vassiliades, K., & Tobet, S. (2011). Gonadotropin-releasing hormone (GnRH) neuron migration: initiation, maintenance and cessation as critical steps to ensure normal reproductive function. *Frontiers in neuroendocrinology*, 32(1), 43–52. https://doi.org/10.1016/j.yfrne.2010.07.005

# Effects of Cross-Sex Hormone Replacement Therapy on Fertility in Transgender Patients: A Systematic Review

## Arin Barth

The following is an excerpt from a longer piece. For full text, please visit https://scholar.colorado.edu/concern/ undergraduate\_honors\_theses/2n49t2555

#### Lay Summary

Although 62% of transgender patients want to have biological kids, 83% of their healthcare providers have never discussed the possible effects of hormone replacement therapy (HRT) on their fertility. Transgender people are often prescribed hormones as a part of gender affirming treatment to change traits that are typically perceived as gendered. Testosterone and estrogen regulate our reproductive systems and impact secondary sex characteristics, such as facial hair and breast development. These hormones can affect one's ability to have kids; however, the extent of that effect is not well researched.

In this paper, I compiled data from 39 studies on the effects of HRT on transgender patients' reproductive systems. I compared and contrasted findings to identify trends and performed statistical analysis to determine the extent of the impact of HRT on the reproductive system. I found that in patients with ovaries, testosterone did not significantly affect the release of egg cells, and only moderately affected the inner layer of the uterus. Meanwhile, estrogen negatively impacted semen quality and therefore fertility. The effect on the reproductive system varied widely from person to person. If HRT was stopped, fertility would generally return to what it was before HRT, but research on long-term effects is especially limited.

Currently, trans people's reproductive choices are limited by lack of research and uninformed healthcare providers, as well as other systemic barriers. This paper provides a comprehensive resource describing our current understanding of this topic. Prioritizing further research in this area will ensure that transgender individuals are empowered to make informed decisions about their reproductive futures.

#### Abstract

A growing area of research among the scientific community is the methods and effects of medical transition for transgender patients. One common component of medical transition is the administration of exogenous cross-sex hormones to patients to obtain a masculinized or feminized phenotype, referred to as hormone replacement therapy (HRT). There is limited research on much of these effects, including the effects of HRT on patients' fertility. In this study, I conducted a systematic review of research on the impacts of HRT on fertility, including calculating risk ratios and etiologic fractions. Thirty-nine papers were examined both qualitatively and quantitatively to assess the impacts of HRT on semen quality, ovulation, polycystic ovary morphology, and endometrial function. This study found that HRT reduces fertility in transgender patients, but not universally or completely; there is still evidence of fertility during HRT administration in many patients. Following discontinuation of HRT, reduced fertility will return to its pre-HRT status in most patients. This research will help clinicians counsel their transgender patients on fertility before and after their medical transition.

#### **Terminology and Abbreviations List**

Assigned female at birth (AFAB): Describes an individual who was assigned a female sex based on their external genitalia at birth.

Assigned male at birth (AMAB): Describes an individual who was assigned a male sex based on their external genitalia at birth.

**Cisgender:** Describes an individual whose gender identity aligns with their assigned sex.

**Estrogen (E):** The primary hormone involved in hormone replacement therapy (HRT) for AMAB individuals. Administration results in a more feminine appearing phenotype for these individuals.

**Gender dysphoria:** An experience of distress or discomfort brought on by a conflict between one's gender identity and sex assigned at birth. Gender dysphoria is experienced by many but not all transgender individuals.

**Gender-nonconforming:** Describes an individual whose behavior or appearance does not conform to prevailing cultural and social expectations about what is appropriate for their gender. These individuals may or may not identify as transgender.

**Hormone replacement therapy (HRT):** The process of administering one or more exogenous hormones to transgender patients to achieve a physical appearance more congruent with their gender identity. One component of medical transition.

**Intersex:** Describes numerous conditions in which sex development does not progress in a manner consistent with typical pathways of "male" or "female." Intersex infants are often coercively assigned a sex at birth, sometimes along with surgical intervention to mimic typical genitalia.

**Medical transition:** The process of undergoing medical treatments to change one's body to better align with one's gender identity. This could include hormone replacement therapy (HRT), voice therapy, laser hair removal, or surgeries such as mastectomy, genital reconstruction surgery, or facial feminization surgery. Many but not all transgender individuals undergo some form of medical transition.

**Nonbinary/genderqueer:** Umbrella terms for individuals whose gender identity falls outside of or is not confined to the established gender binary. These terms fall under the "transgender" umbrella, but there are nuanced social, political, and cultural differences between them, and they are not interchangeable.

**Testosterone (T):** The primary hormone involved in hormone replacement therapy (HRT) for AFAB individuals. Administration results in a more masculine appearing phenotype for these individuals.

**Transgender:** Describes an individual whose gender identity differs from their assigned sex.

**Transgender men:** Individuals whose gender identity is male and who were assigned female at birth (AFAB). **Transgender women:** Individuals whose gender identity is female and who were assigned male at birth (AMAB).

#### Introduction

Medical care for transgender individuals is a growing field in health care as broad cultural awareness and acceptance of diverse gender identity increases.... One common component of medical transition is hormone replacement therapy (HRT), where one or more exogenous sex hormones are administered to change a patient's hormone profile and therefore, the phenotype of some secondary sex characteristics.

#### [...]

Many of the full-body effects of HRT are moderately well-researched and understood; however, research on the effects of HRT on fertility is lacking.

#### [...]

In AMAB individuals taking HRT, high serum estrogen could result in reduced nutrients, growth factors, and

proteins needed for proper sperm development and impaired release of mature spermatids (Figure 1c).

[...]



Figure 1: Diagram of the typical HPG axis (a), an HPG axis with interference by supraphysiological exogenous testosterone (b), and an HPG axis with interference by supraphysiological exogenous estrogen (c). Arrows between boxes indication stimulation and (-) indicates feedback regulation. Small arrows indicate increased or decreased hormone levels relative to baseline. Bold lines indicate increased strength of negative feedback.
Conversely, high serum testosterone in AFAB patients is less likely to disrupt the hormone cycles necessary for ovulation because of the intricacies of the feedback mechanisms (Figure 1b).

[...]

Overall, I hypothesize measures of fertility in transgender patients on HRT to be significantly reduced relative to cisgender controls, but not indicating a complete absence of fertility among the population.

## Methods

#### Literature Search

two in AMAB subjects.

A literature review of the biomedical database PubMed was performed from August 28, 2019 to October 17, 2019.

[...]

Data Analysis Risk Ratios (RR) and confidence intervals (CI) were calculated for eight studies: six in AFAB subjects, and

## [...]

Standard errors (SE) were calculated from CIs ....

The etiologic fraction (EF) indicates the proportion of the changes to a specific outcome that can be attributed to the treatment applied, which is HRT for the purposes of this study.

## [...]

For outcomes which had a RR less than one, the EF calculated by this method is not meaningful, so an alternative etiologic fraction was (AEF) was calculated.

[...]

Prevalence rates were calculated for various outcomes in all patients taking HRT in all studies: presence of a corpus lutea or corpus albicantia, endometrial activity, presence of endometrial polyps, polycystic ovary morphology, and spermatogenesis.

[...]

## Results

#### Literature Search Results

During the literature search, 567 articles were screened for relevancy. Two hundred and twenty studies in non-humans were excluded and 347 human studies were screened for relevancy. Of these, 39 articles were determined to meet the criteria and were relevant to the research.

[...]

## Results in AMAB Patients

Twenty relevant articles in AMAB patients were examined for this analysis (Table 1). Risk ratios, 95% confidence intervals, and standard errors were calculated for two papers in transgender women (Figure 3, Table 2). Data in this section is listed as "RR (Lower CI-Upper CI, SE)."

All of the RRs I calculated were greater than 1, which indicated <u>increased</u> risk of sperm abnormalities in transgender patients relative to cisgender controls.

# 

📄 Complete 🛑 Reduced 🔵 Absent

Figure 4: Calculated prevalence rates of observed complete, reduced, or absent spermatogenesis in the testes of AMAB transgender people taking HRT.

## [...]

Results in AFAB Patients

Nineteen articles in AFAB patients were examined for this analysis. Risk ratios, 95% confidence intervals, and standard errors were calculated for six papers in AFAB patients (Figure 5, Tables 4 and 5).

Several outcomes resulted in RRs greater than 1, indicating <u>increased</u> risk of the relevant outcome in transgender patients relative to cisgender controls.

[...]

Several other outcomes had RRs less than or equal to 1, including several studies noting the presence of a corpus lutea and/or albicantia.

[...]

η'n m năn năn năn η'n năn năn năn nîn nîn năn năn năn năn năn năn năn din din **n**p nin. nîn nîn nn, nn. nn. n p ŵ ŵ ŵ ŵ nîn nîn n n n nă nă nă m nin. ΠÌΠ. nin. m. nn. η**η** nîn nîn nîn nîn nin. Ť Ť Ť nîn nîn ΠÌ. m nîn nîn ŵ ŵ ŵ ŵ ŵ ŵ ŵ ŵ η'n. nîn. η'n. η'n. η Π (III) η**η** nin nin nîn nîn ΠÌ η**η** m

🔵 Corpus luteum and/or albicantia present 🕘 Neither present

a.

ŵŵŵŵŵŵŵŵŵŵ ŵŵŵŵŵŵŵŵŵŵŵŵŵŵ m m m ŵ ŵ nîn. m, m m m m η'n. η'n nîn nîn nin. nîn. nn. nn nn, **m** ŵ ŵ ŵ ŵ ŵ ŵ ŵ ŵ năn năn năn η'n η'n. nîn nîn η'n η**p** 

🛑 PCOM 🔵 No PCOM

b.

ŵ ŵ ŵ ŵ nh nh nh nh nin nin n n n ŵ ŵ ŵ η'n. ΠÌ. nin. nîn nîn n ŵ ŵ m **n**in m, nîn. m nn. nn. m nn i **m** m nn. **m m** n n nn. nn. m nn. nn, m m **m U** η**n** η'n ΠÌ. **m** <u>ݰ</u>ݑݰ**ݰ**ݰ**ݰ**ݰ**ݰ**ݰ**ݰ**ݰ<u></u>ݰ<u></u>ݰ</u><u>ݰ</u><u></u>ݰ</u><u></u><u></u><u></u><u></u><u></u><u></u>

Secretory Proliferative Inactive Atrophied

c.

Figure 6: Calculated prevalence rates of evidence of ovulation (a), PCOM (b), and endometrial states (c) observed AFAB transgender people taking HRT.

#### Discussion

#### Effects in AMAB Patients

Although there is some disagreement between studies, semen quality in AMAB transgender patients currently using HRT appears to be significantly lower than in cisgender controls. Transgender patients were shown to have a statistically significant increased risk of oligospermia (decreased sperm concentration) and of teratospermia (abnormal sperm morphology) compared to cisgender controls.

[...]

#### Effects in AFAB Patients

One of the most important factors in whether an AFAB individual can conceive is the presence or absence of ovulation. Evidence of ovulation in AFAB transgender individuals taking hormone replacement therapy were not necessarily significantly reduced.... Taken together, these data indicate that HRT does not necessarily negatively impact ovulation in AFAB patients.

#### [...]

Hormone replacement therapy did not appear to result in endometrial atrophy in most AFAB patients.

#### [...]

Effects Following HRT Discontinuation Following discontinuation of HRT in AMAB patients, sperm quality seems to largely return to pre-HRT levels. ... These data indicate that semen quality return to baseline levels according to most measures in as little as 6 weeks post discontinuation of HRT in AMAB patients.

There is limited data on the fertility of AFAB patients following discontinuation of HRT. ...

Hormone replacement therapy in AFAB patients does not result in long-term infertility.

[...]

#### Non-hormonal Factors

It is important to consider factors besides HRT that could result in reduced fertility in transgender patients. It was not possible for this analysis to control for possible confounding variables, but differences in fertility before initiation of HRT are important to consider. Many studies recorded baseline measures for their subjects, otherwise assessed pre-treatment parameters, or studied patients prior to initiating HRT.

[...]

#### Strengths and Limitations

This is the first broad systematic review of the effects of HRT on fertility in both AFAB and AMAB subjects. However, this study was limited in several ways. First, I was unable to perform meta-analysis. Relatively few studies met all the criteria for this type of analysis, because of failure to use controls, fully discuss the results of subjects in control groups, or because the sample sizes were too small to provide adequate data for a defined risk ratio. If a meta-analysis had been possible, it would have lent another dimension of statistical analysis to the data, improving the overall understanding of the overall findings of the research available.

#### [...]

This review of existing research makes the scientific understanding of fertility on HRT more accessible to health care providers, so that they can accurately counsel their patients on how HRT may affect their fertility, so that patients can more easily make informed decisions about their bodies.

Additionally, this is the only study that I know of that focuses on the effects of HRT, or any other aspect of medical transition, on transgender patients, that was authored by a transgender researcher.

[...]

Research into transgender bodies is not complete without an understanding of transgender communities....

#### Conclusions

There is a shortage of data on the effects of HRT on the fertility of transgender individuals, and what data does exist is often conflicting or inconclusive. The data we have indicates that fertility may be somewhat, but not completely reduced in transgender patients. Effects of HRT do not appear to increase past initial changes to cause more significant changes with increased duration, and HRT does not result in long term loss of fertility; following discontinuation of the medication normal fertility should return in most patients. However, there is some possibility that reduction in semen quality in AMAB transgender patients could be present prior to HRT, and therefore related to other factors such as stress and anxiety. Transgender patients hoping to avoid pregnancy should use birth control methods and not consider their HRT regimen to be a substitute for birth control. Transgender patients hoping to conceive should be able to do so but may wish to consider discontinuing HRT to raise their chances. Transgender patients who were assigned female at birth will have to discontinue HRT use to sustain a healthy pregnancy.

#### Works Cited

- Adeleye, A. J., Reid, G., Kao, C. N., Mok-Lin, E., & Smith, J. F. (2019). Semen Parameters Among Transgender Women With a History of Hormonal Treatment. *Urology*, 124, 136–141. https://doi.org/10.1016/j.urology.2018.10.005
- Adkins, B. D., Barlow, A. B., Jack, A., Schultenover, S. J., Desouki, M. M., Coogan, A. C., & Weiss, V. L. (2018). Characteristic findings of cervical Papanicolaou tests from transgender patients on androgen therapy: Challenges in detecting dysplasia. *Cytopathology*, 29(3), 281–287. https://doi.org/10.1111/cyt.12525
- Amirikia, H., Savoy-Moore, R. T., Sundareson, A. S., & Moghissi, K. S. (1986). The effects of long-term androgen treatment on the ovary. *Fertility and Sterility*, 45(2), 202–208. https://doi.org/10.1016/S0015-0282(16)49155-7
- Barnard, E. P., Dhar, C. P., Rothenberg, S. S., Menke, M. N., Witchel, S. F., Montano, G. T., ... Valli-Pulaski, H. (2019). Fertility Preservation Outcomes in Adolescent and Young Adult Feminizing Transgender Patients. *Pediatrics*, 144(3), e20183943. https://doi.org/10.1542/peds.2018-3943
- Caanen, M. R., Schouten, N. E., Kuijper, E. A. M., Van Rijswijk, J., Van Den Berg, M. H., Van Dulmen-Den Broeder, E., ... Lambalk, C. B. (2017). Effects of long-term exogenous testosterone administration on ovarian morphology, determined by transvaginal (3D) ultrasound in female-to-male transsexuals. *Human Reproduction*, 32(7), 1457–1464. https://doi.org/10.1093/humrep/dexo98
- Chen, D., Matson, M., Macapagal, K., Johnson, E. K., Rosoklija, I., Finlayson, C., ... Mustanski, B. (2018). Attitudes Toward Fertility and Reproductive Health Among Transgender and Gender-Nonconforming Adolescents. *Journal of Adolescent Health*, 63(1), 62–68. https://doi.org/10.1016/j.jadohealth.2017.11.306
- Chiu, A., Chen, M.-T., Chiang, H., Wu, L.-H., Fang, R.-H., & Chang, L. (1993). Deoxyribonucleic Acid Histogram of Testes in Primary Transsexualism. *British Journal of Urology*, 72, 495–497. https://doi.org/10.1039/c2cs35296f
- De Roo, Chloë, Lierman, S., Tilleman, K., Peynshaert, K., Braeckmans, K., Caanen, M., ... De Sutter, P. (2017). Ovarian tissue cryopreservation in female-to-male transgender people: insights into ovarian histology and physiology after prolonged androgen treatment. *Reproductive BioMedicine Online*, *34*(6), 557–566. https://doi.org/10.1016/j.rbm0.2017.03.008
- De Roo, Chloe, Tilleman, K., Vercruysse, C., Declercq, H., T'Sjoen, G., Weyers, S., & De Sutter, P. (2019). Texture profile analysis reveals a stiffer ovarian cortex after testosterone therapy: a pilot study. *Journal of Assisted Reproduction and Genetics*. https://doi.org/10.1007/s10815-019-01513-x
- Durairajanayagam, D. (2018). Lifestyle causes of male infertility. *Arab Journal of Urology*, *16*(1), 10–20. https://doi.org/10.1016/j. aju.2017.12.004

- Flores, A. R., Herman, J. L., Gates, G. J., & Brown, T. N. T. (2016). How Many Adults Identify As Transgender in the United States? *The Williams Institute*, (June), 13. Retrieved from https://williamsinstitute.law.ucla.edu/wp-content/uploads/How-Many-Adults-Identify-as-Transgender-in-the-United-States.pdf
- Futterweit, W., & Deligdisch, L. (1985). Histopathological Effects of exogenously administered testosterone in 19 female to male transsexuals. *Journal of Chemical Information and Metabolism*, 62(1), 16–21. https://doi.org/10.1017/CBO9781107415324.004
- Gooren, L. J. G. (1985). Exogenous androgens decrease the length of the luteal phase and increase the length of the follicular phase. *Hormone and Metabolic Research*, 17(12), 683–684.
- Gooren, L. J. G., & Giltay, E. J. (2008). Review of studies of androgen treatment of female-to-male transsexuals: Effects and risks of administration of androgens to females. *Journal of Sexual Medicine*, 5(4), 765–776. https://doi.org/10.1111/j.1743-6109.2007.00646.x
- Grimstad, F. W., Fowler, K. G., New, E. P., Ferrando, C. A., Pollard, R. R., Chapman, G., ... Gray, M. (2019). Uterine pathology in transmasculine persons on testosterone: a retrospective multicenter case series. *American Journal of Obstetrics and Gynecology*, 220(3), 257.e1-257.e7. https://doi.org/10.1016/j.ajog.2018.12.021
- Grynberg, M., Fanchin, R., Dubost, G., Colau, J. C., Brémont-Weil, C., Frydman, R., & Ayoubi, J. M. (2010). Histology of genital tract and breast tissue after long-term testosterone administration in a female-to-male transsexual population. *Reproductive BioMedicine Online*, 20(4), 553–558. https://doi.org/10.1016/j.rbm0.2009.12.021
- Hamada, A., Kingsberg, S., Wierckx, K., T'Sjoen, G., De Sutter, P., Knudson, G., & Agarwal, A. (2015). Semen characteristics of transwomen referred for sperm banking before sex transition: A case series. *Andrologia*, 47(7), 832–838. https://doi.org/10.1111/and.12330
- Ikeda, K., Baba, T., Noguchi, H., Nagasawa, K., Endo, T., Kiya, T., & Saito, T. (2013). Excessive androgen exposure in female-to-male transsexual persons of reproductive age induces hyperplasia of the ovarian cortex and stroma but not polycystic ovary morphology. *Human Reproduction*, 28(2), 453–461. https://doi.org/10.1093/humrep/des385
- Jindarak, S., Nilprapha, K., Atikankul, T., Angspatt, A., Pungrasmi, P., Iamphongsai, S., ... Tiewtranon, P. (2018). Spermatogenesis abnormalities following hormonal therapy in transwomen. *BioMed Research International*, 2018. https://doi.org/10.1155/2018/7919481
- Jurewicz, J., Radwan, M., Sobala, W., Radwan, P., Jakubowski, L., Hawuła, W., ... Hanke, W. (2014). Lifestyle factors and sperm aneuploidy. *Reproductive Biology*, 14(3), 190–199. https://doi.org/10.1016/j.repbio.2014.02.002
- Kent, M. A., Winoker, J. S., & Grotas, A. B. (2018). Effects of Feminizing Hormones on Sperm Production and Malignant Changes: Microscopic Examination of Post Orchiectomy Specimens in Transwomen. Urology, 121, 93–96. https://doi.org/10.1016/j.urology.2018.07.023
- Khalifa, M. A., Toyama, A., Klein, M. E., & Santiago, V. (2018). Histologic Features of Hysterectomy Specimens from Female-Male Transgender Individuals. *International Journal of Gynecological Pathology*, (5), 1–8. https://doi.org/10.1097/PGP.000000000000548
- Kisman, O. K., De Voogt, H. J., & Baak, J. A. (1990). Reversibility of the effect of LHRH agonists and other antiandrogenic hormones on the testis: A histomorphometric study. *European Urology*, *18*(4), 299–301. https://doi.org/10.1159/000463932
- Li, K., Rodriguez, D., Gabrielsen, J. S., Centola, G. M., & Tanrikut, C. (2018). Sperm cryopreservation of transgender individuals: trends and findings in the past decade. *Andrology*, 6(6), 860–864. https://doi.org/10.1111/andr.12527
- Lierman, S., Tilleman, K., Braeckmans, K., Peynshaert, K., Weyers, S., T'Sjoen, G., & De Sutter, P. (2017). Fertility preservation for trans men: frozen-thawed in vitro matured oocytes collected at the time of ovarian tissue processing exhibit normal meiotic spindles. *Journal* of Assisted Reproduction and Genetics, 34(11), 1449–1456. https://doi.org/10.1007/s10815-017-0976-5
- Light, A. D., Obedin-Maliver, J., Sevelius, J. M., & Kerns, J. L. (2014). Transgender men who experienced pregnancy after female-to-male gender transitioning. *Obstetrics and Gynecology*, 124(6), 1120–1127. https://doi.org/10.1097/AOG.00000000000540
- Loverro, G., Resta, L., Dellino, M., Edoardo, D. N., Cascarano, M. A., Loverro, M., & Mastrolia, S. A. (2016). Uterine and ovarian changes during testosterone administration in young female-to-male transsexuals. *Taiwanese Journal of Obstetrics and Gynecology*, 55(5), 686–691. https://doi.org/10.1016/j.tjog.2016.03.004
- Lu, C., & Steinberger, A. (1978). Effects of estrogen on human seminferous tubules: light and electron microscope analysis. *The American Journal of Anatomy*, 153(1), 1–13.
- Lubbert, H., Leo-Rossberg, I., & Hammerstein, J. (1992). Lubbert H- Effects of ethinyl estradiol on semen quality and various hormonal parameters in a eugonadal male. (Fertil Steril 58-603, 1992).pdf. 58(3), 603–608.
- Marsh, C., McCracken, M., Gray, M., Nangia, A., Gay, J., & Roby, K. F. (2019). Low total motile sperm in transgender women seeking hormone therapy. *Journal of Assisted Reproduction and Genetics*, 1639–1648. https://doi.org/10.1007/s10815-019-01504-y
- Matoso, A., Khandakar, B., Yuan, S., Wu, T., Wang, L. J., Lombardo, K. A., ... Yakirevich, E. (2018). Spectrum of findings in orchiectomy specimens of persons undergoing gender confirmation surgery. *Human Pathology*, 76, 91–99. https://doi.org/10.1016/j.humpath.2018.03.007

- Melmed, S., Polonsky, K. S., Larsen, P. R., & Kronenberg, H. M. (2012). *William's Textbook of Endocrinology* (12th ed.; P. Hetherington & J. Ryan, eds.). Philadephia: Elsevier Inc.
- Meriggiola, M. C., & Gava, G. (2015a). Endocrine care of transpeople part I. A review of cross-sex hormonal treatments, outcomes and adverse effects in transmen. *Clinical Endocrinology*, 597–606. https://doi.org/10.1111/cen.12753
- Meriggiola, M. C., & Gava, G. (2015b). Endocrine care of transpeople part II. A review of cross-sex hormonal treatments, outcomes and adverse effects in transwomen. *Clinical Endocrinology*, *8*<sub>3</sub>(5), 607–615. https://doi.org/10.1111/cen.12754
- Miller, N., Bédard, Y. C., Cooter, N. B., & Shaul, D. L. (1986). Histological changes in the genital tract in transsexual women following androgen therapy. *Histopathology*, 10(7), 661–669. https://doi.org/10.1111/j.1365-2559.1986.tbo2520.x
- Nistal, M., Gonzalez-Peramato, P., & De Miguel, M. P. (2013). Sertoli cell dedifferentiation in human cryptorchidism and gender reassignment shows similarities between fetal environmental and adult medical treatment estrogen and antiandrogen exposure. *Reproductive Toxicology*, 42, 172–179. https://doi.org/10.1016/j.reprotox.2013.08.009
- Pache, T. D., Chadha, S., Gooren, L. J. G., Hop, W. C. J., Jaarsma, K. W., Dommerholt, H. B. R., & Fauser, B. C. J. M. (1991). Ovarian morphology in long-term androgen-treated female to male transsexuals. A human model for the study of polycystic ovarian syndrome? *Histopathology*, 19(5), 445–452. https://doi.org/10.1111/j.1365-2559.1991.tb00235.x
- Pache, T. D., Hopt, W. C. J., De Jong, F., Leerentveld, R., van Geldorp, H., Van de Kamp, T., ... FAUSER, B. C. J. M. (1992). 17B-Oestradiol, androstenedione and inhibin levels in fluid from individual follicles of normal and polycystic ovaries, and in ovaries from androgen treated female to male transexuals. *Clinical Endocrinology*, 36(6), 565–571. https://doi.org/10.1017/CBO9781107415324.004
- Payer, A. F., Meyer, W. J., & Walker, P. A. (1979). The Ultrastructural Response of Human Leydig Cells to Exogenous Estrogens. Andrologia, tt(6), 423–436. https://doi.org/10.1111/j.1439-0272.1979.tb02232.x
- Peitzmeier, S. M., Reisner, S. L., Harigopal, P., & Potter, J. (2014). Female-to-male patients have high prevalence of unsatisfactory paps compared to non-transgender females: Implications for cervical cancer screening. *Journal of General Internal Medicine*, 29(5), 778–784. https://doi.org/10.1007/s11606-013-2753-1
- Perrone, A. M., Cerpolini, S., Maria Salfi, N. C., Ceccarelli, C., De Giorgi, L. B., Formelli, G., ... Meriggiola, M. C. (2009). Effect of long-term testosterone administration on the endometrium of female-to-male (FtM) transsexuals. *Journal of Sexual Medicine*, 6(11), 3193–3200. https://doi.org/10.1111/j.1743-6109.2009.01380.x
- Rodriguez-Rigau, L. J., Tcholakian, R. K., Smith, D., & Steinberger, E. (1977). In vitro steroid metabolic studies in human testes II: metabolism of cholesterol, pregnenolone, progesterone, androstendione and testosterone by testes of an estrogen-treated man. *Steroids*.
- Sapino, A., Pagani, A., Codano, A., & Bussolati, G. (1987). Effects of estrogens on the testis of transsexuals: A pathological and immunocytochemical study. *Virchows Archiv A Pathological Anatomy and Histopathology*, 411(5), 409–414. https://doi.org/10.1007/ BF00735221
- Schneider, F., Neuhaus, N., Wistuba, J., Zitzmann, M., He
  ß, J., Mahler, D., ... Kliesch, S. (2015a). Testicular Functions and Clinical Characterization of Patients with Gender Dysphoria (GD) Undergoing Sex Reassignment Surgery (SRS). *Journal of Sexual Medicine*, 12(11), 2190–2200. https://doi.org/10.1111/jsm.13022
- Schneider, F., Neuhaus, N., Wistuba, J., Zitzmann, M., He
  ß, J., Mahler, D., ... Kliesch, S. (2015b). Testicular Functions and Clinical Characterization of Patients with Gender Dysphoria (GD) Undergoing Sex Reassignment Surgery (SRS). *Journal of Sexual Medicine*, 12(11), 2190–2200. https://doi.org/10.1111/jsm.13022
- Schulze, C. (1988). Response of the human testis to long-term estrogen treatment: Morphology of Sertoli cells, Leydig cells and spermatogonial stem cells. *Cell and Tissue Research*, 251(1), 31–43. https://doi.org/10.1007/BF00215444
- Spinder, T., Spijkstra, J. J., Van Den Tweel, J. G., Burger, C. W., Van Kessel, H., Hompes, P. G. A., & Gooren, L. J. G. (1989). The effects of long term testosterone administration on pulsatile luteinizing hormone secretion and on ovarian histology in eugonadal female to male transsexual subjects. *Journal of Clinical Endocrinology and Metabolism*, 69(1), 151–157. https://doi.org/10.1210/jcem-69-1-151
- Thiagaraj, D., Gunasegaram, R., Loganath, A., Peh, K., Kottegoda, S., & RAtnam, S. (1987). *Histopathology of the Testes from Male Transexuals on Oestrogen Therapy*.
- Venizelos, I., & Paradinas, F. (1987). Brief report Testicular atrophy after oestrogen therapy. 1, 3-6.

# Little Ambergris Cay, a Case Study for Ooid Rich Island Development on the Turks and Caicos Carbonate Platform

## Drew G. Brown

The following is an excerpt from a longer piece. For full text, please visit https://scholar.colorado.edu/concern/undergraduate\_honors\_theses/8910jv661

#### Lay Summary:

This study proposes a hypothesis for the formation of an island located within the Turks and Caicos carbonate platform, named Little Ambergris Cay. Little Ambergris Cay is unusual in that it formed during a period of sea level rise within the Holocene. This anomaly coincided with unique sediment characteristics, as the island is dominantly composed of ooids. Ooids are concentrically coated calcium carbonate sand grains. In this circumstance, the rapid accumulation of ooids resulted in island formation. This coupling of high ooid concentrations alongside sea level rise provided an opportunity to build a new model for modern carbonate island growth. This study used data from the loose sediments of Little Ambergris Cay, such as grain type, grain size, biological composition, and roundness of grains, compared with different carbonate islands data, allowing us to track changes in environment over time. By performing this comparison, we can see how Little Ambergris Cay formed within the previous five thousand years. Through interpreting the changes in environment over this time, we draw conclusions about formation direction and the possible reasons for accumulation during sea level rise. This study provides a unique model for carbonate island formation that has relevance in the carbonate sedimentology field.

#### Abstract:

Little Ambergris Cay (LAC) within the Turks and

Caicos is a useful field site to examine the accumulation of a carbonate island, within a typical carbonate platform environment, because it does not follow the typically accepted modes of island accretion. There are several methods used to describe carbonate island formation after Schlager (2003), who described carbonate "factories" in reference to the modern analogues of Florida and the Bahamas. The Turks and Caicos, and specifically Little Ambergris Cay, offers a field side that differs from the comparison analogues in sediment composition, energy flux, and Holocene development. Little Ambergris Cay is unusual in that it formed during a period of sea level rise (Figure 2, Toscano & Macintyre, 2003), and thus presents an opportunity for research on how modern carbonate platform environments accumulate through time. This paper presents an argument for eastward direction of island accumulation of Little Ambergris Cay. Based on the research of Trower et al. (2018), Dravis & Wanless (2008), and Schlager (2003), I interpret that Little Ambergris Cay formed via eastward accumulation based on the influence of wind-wave current energy as means of supplying ample sediment to a zone of accumulation; in other words, in this system, sediment supply outpaces accommodation space. To support this interpretation, I will present several sets of data that were derived from unlithified sediment cores collected from the interior of LAC as means to characterize and compare against representative environments. First, I performed thin section point-count analysis on loose

sediments from a variety of depths in a representative collected core to identify how grain composition varied with depth. I then compared this point count data with a range of grain characteristics such as grain size, sorting, sphericity, etc., determined via Camsizer analysis, in order to determine the parameters that best described the changes in grain concentration as seen in the point count. Finally, by integrating plots of the grain size parameters determined in this paper vs. depth, stereoscope microscopy, and radiocarbon ages of core sediments, I compared representative facies of modern carbonate environments to determine a facies development through time of Little Ambergris Cay. I concluded that Little Ambergris Cay underwent a series of high to low energy facies changes, where early lithification on the south-western edge of the modern LAC allowed successive accumulation of incoming easterly ooids due to sediment entrapment, possibly due to microbial mat and bedrock influence. The incoming sediment grains from the east onto an early lithified fragment to the west allowed eastward accumulation, where facies changed from an ancient shoal to a modern island. This paper furthers the research of modern carbonate analogues and their interpreted growth patterns. LAC offers a unique environment that does not follow typical carbonate island development, and by providing a case study of wind-wave influenced island accretion, sectors of geologic interest have the opportunity to interpret unique climatic and chemical factors. The petroleum unconventional subsurface carbonate play search is greatly related to LAC, as grainstones offer potential reservoirs (Dravis & Wanless, 2008). The geochemical field may have interest in the accumulation of LAC, as specific seawater concentrations play a role in the uniquely high concentrations of ooids found on the Caicos Platform. The geobiological field may be interested in LAC, as the relatively low abundance of microbes and biotic influence differs greatly from the other modern analogues for carbonate platform

development.

#### Introduction:

Little Ambergris Cay (LAC, Figure 1) within the Turks and Caicos is known to be composed of dominantly Holocene aged carbonate sediments (Orzechowski et al., 2016). Toscano & Macintyre (2003) recognized that a period of sea level rise occurred in the Atlantic Holocene. Generally speaking, using the Bahamas and Florida as a modern analogue for carbonate platform development, the most common way for carbonate accumulation to outpace or remain at increasing sea level is through a process of "backstepping," or retrogradation of the platform that is primarily influenced by reef growth (Schlager, 2003). Reefs are rare on the Caicos platform, so considering that we instead primarily observe Holocene ooids (Dravis and Wanless, 1989), and recognizing that LAC accreted past sea level during a period of increasing accommodation space, it then becomes clear that LAC does not follow the most obvious modern models of island formation in the Holocene. This anomaly offers an opportunity to better understand how and why formation of this carbonate platform island occurred in such conditions.

## [...]

Before research was conducted, I developed hypotheses regarding the question of how an island forms during a period of sea level rise. I hypothesized that, in order for an island to form, sediment supply must have outpaced sea level rise (accommodation space). Trower et al. (2018), proposed a westward growth hypothesis of LAC, which I decided to test through facies development and stratigraphy. It is known that current and wind influence westward migration of sediment (Trower et al., 2018; Dravis and Wanless, 2008), and Big Ambergris Cay lies to the east of LAC and is



Figure 1: A: A map of Little Ambergris Cay with associated core locations.

older in age. Given this observation, I hypothesize that sediments from BAC are migrating to LAC where early lithification of ooid grainstones could have trapped incoming sediments. I therefore also developed an alternative hypothesis that Little Ambergris Cay accreted eastward, towards BAC, as a function of incoming sediment supply onto a pre-existing bedrock.

Three steps will be taken in testing these hypotheses (westward vs eastward accretion), by analyzing sediment sample characteristics from cores collected on the interior of LAC. Using point counting data (Figure 3), I am able to determine sediment composition within the VCo3 core. Secondly, I used Matlab to analyze Camsizer data, and was able to determine characteristic grain size (D10, D50, D90), mean roundness, and sorting for the other cores (VC01-VC08). By comparing the point counting data to the Camsizer data for VCo<sub>3</sub>, I determined that the roundness, grain size and sorting parameters correlated to the interpreted grain composition.... The third and final step was to use both the point counting and Camsizer data, compared against a set of thin sections and Camsizer data, for samples that represented

modern depositional environments to determine facies and environmental change over time on LAC.

Based on the conclusions that I draw in these three steps, using the parameters described above, I will test that the westward growth hypothesis of Little Ambergris Cay (Trower et al., 2018) is not supported by the data analyzed in this paper. Rather, based on the facies development of the VC cores and the comparison facies deposits of Table 1 & 2, I will attempt to support an eastward growth hypothesis of Little Ambergris Cay.

#### **Materials and Methods:**

#### [...]

Sediment Core and Sample Collection I began work on a set of sediment samples collected from a variety of depths in eight sediment cores previously collected in 2017 and 2018 via vibracoring.

[...]

#### Point Count Analysis Data

The first step was to conduct a point count for sediment composition using JMicrovision, an image analysis toolbox. Analyzing 11 thin sections of the loose sediment in VCo3, at each successive depth (up to 220cm), I performed a 400-sample point count for each increment depth.

[...]

Grain Size and Shape Data

The next step was to analyze grain size and shape data, which had previously been measured via a Retsch Camsizer P<sub>4</sub>, from all the cores through a Matlab code that was able to process the  $D_{10}$ ,  $D_{50}$ ,  $D_{50}$ , mean roundness, mean sphericity, mean aspect ratio, roundness, and sorting.

[...]

Stereoscope Microscopy

Stereoscope microscopy was performed on the four samples.... This imaging allowed me to qualitatively assess the abundance of ooids and skeletal grains, and confirm the Camsizer sorting, roundness, and diameter data.

#### Radiocarbon Data

Radiocarbon dating was performed on 11 samples (Figure 2 & FA-2-A and FA-4-A) via Accelerator Mass Spectrometer (AMS) at the Keck-Carbon Cycle AMS facility at UC Irvine and the NOSAMS facility at WHOI.

[...]

Facies Correlations to Modern Depositional Environments

...By collecting the sorting and roundness data for the representative environments, I was able to compare the previously processed sorting and roundness data of the VC cores and correlate each depth's values to the associated representative environments. This allowed me, along with the radiocarbon age of the given cores, to formulate an idea of how facies changed over time.

#### **Results:**

The ages of the three cores (ranging from youngest to oldest) display similar trends vs. depth between each core. CC-J18 and VCo3 are in similar positions on the northern side of LAC, while FML-J18A is located on the eastern side of LAC (Figure 1.A) Corrected radiocarbon age for exposed bedrock samples FA-2-A & FA-4-A (Figure 1.C) show ages of 5946BP & 5220BP. The bedrock has been interpreted as a crossstratified eolianite ooid grainstone (Orzechowski et al. 2016).



Figure 2: Corrected radiocarbon data for three cores collected on northern LAC.



Figure 3: The scatter plot above shows sediment compositions a function of core depth.

The data was determined through a 400-sample point count for each depth in VCo3. Trends imply that skeletal grains were abundant in the shallower and deeper sections of the core, and ooid abundance increased in the mid-core (60cm+) and decreased in the late-core (~150cm), followed by another period of increase (~180cm). Sediment composition provides key information into depositional environment and will be discussed in more specific analysis further in the discussion. Each depth displayed offers three general trends: percent changes in ooid, peloid, and skeletal grain abundance. I determined that roundness, median diameter (D50), and sorting, as provided through Camsizer analysis, best correlated with the changes in concentration of ooid and skeletal presence described in Figure 3, and should thus be interpreted for other cores of interest as described in Figures 4, 5 and 6.



Figure 4: The photos above represent each core's depth compared against roundness. The higher the values, the higher the roundness.

Four cores were chosen to compare roundness, D50, and sorting versus depth to serve as a proxy for ooid concentration. Each core is located within different cardinal direction points around LAC (Figure 1.A) to give an interpretation of ooid concentrations based on differing locations.

The maximum roundness described in Figure 4.B is 0.81 in VC03, while the minimum value is 0.38 in VC08. Each core shows different patterns vs. depth, where VC08 increases with depth, and VC03 initially increases, but gradually decreases following 100cm. VCo<sub>4</sub> and VCo1 remain relatively constant throughout their entire section.

The maximum median diameter within the described cores is just <1200m in VC03. The minimum median diameter is ~300m in VC08. Once again, VC04 and VC01 display relatively consistent values throughout their section. Median diameter gives an interpretation of overall grain size, where consistent values of D50 represent consistent grain composition



Figure 5: The photos above represent each core's depth compared against D50, or the median diameter. Units of D50 are in microns.



Figure 6: The figures above represent each core's depth compared against sorting. Higher numbers reflect poorer sorting.

The lowest degree of sorting is described in VCo8 at  $\sim$  5.5. The best degree of sorting is also within VCo8 at 0.35 at 80cm depth. VCo3 also exhibits a high degree of sorting at 0.38 at 60cm depth. Sorting provides an interpretation of grain presence, where the best degrees of sorting for LAC are consisting of dominantly ooids, and the lower degrees are a large mix of skeletal grains with ooids..



Figure 7:

The three photographs above show the interpretation of the succession of depositional facies through time based on the radiocarbon ooid data collected (Figure 2).

#### **Discussion:**

[...]

Regarding the question of why LAC accumulates in the position it does (~3 miles off of BAC), instead of accreting off the western edge of BAC, I have formulated three hypotheses that could explain such formation.

1. BAC poses as a large geographic "block" for current and wind-wave influence coming in from the east. This "block" forces wave refraction off the northern and southern edges of BAC. It is possible that the two refractions combine at the  $\sim 3$  miles mark where the initial bedrock could have lithified, thus allowing sediment supply to be particularly higher at this specific location, and in turn allowing eolianite processes or cement compaction processes to allow an early lithification.

2. Patch reefs are visible and have been observed around the entirety of the Caicos Platform (Dravis and Wanless, 2008). It is possible that a singular patch reef was/is in-situ at the  $\sim$  3-mile point west of BAC. It is possible that the westward migrating ooids were able to be trapped in this patch reef, and in turn allow sediment accumulation to form an early lithification to surpass sea level rise.

3. Knowing that ooids are increasing in size as they transport west (Trower et al., 2018), it could be possible that the Intra-Cay shoal grew westward enough to sufficiently surpass water depth. In other words, it could be possible that the westward growth of ooids along the BAC shoal grew to high enough size in high enough concentrations that they were able to succeed sea level, and thus allow an early lithification along with eolianite processes. i. Microbial mat influence has the potential to affect each of the above listed scenarios. While in-situ biology has not proved to be a leading cause of LAC island accumulation based on the research provided in this paper, it should be noted that microbial mats greatly increase the ability for sediment entrapment, and even minor presence of microbial mats could help support accumulation of incoming ooid sediments.

#### Future Directions

...However, I propose that research be conducted on a series of processes within the Intra-Cay shoal and LAC. Based on the above hypotheses, the direction and energy relationship of refraction off BAC should be considered in how exactly the wave energy may influence the combination or direction of ooid sediment transport.

[...]

#### **Conclusion:**

*Building the Overall Picture of Little Ambergris Cay Formation* 

1. The Ambergris Shoal (Intra-Cay Shoal), extending a total of ~2.2km west (reaching the modern LAC shoreline) of BAC, of age greater than 6000BP, develops due to easterly wind-wave action and the prior abundance of ooid grains.

2. An abundant supply of ooids are continuously transported from BAC to the west, accumulating and extending along the Intra-Cay shoal.

3. Sediment supply outpaces accommodation space following the eolianite influence of LAC ooid bedrock lithification  $\sim 5946$ BP. Possible reasons for sediment supply to accumulate past the subsurface and lithify could be due to wave refraction induced sediment accumulation, patch reef presence and subsequent sediment accumulation, or transport induced ooid growth resulting in carbonate sediment growth surpassing sea level rise.

4. Westward migration of ooids became trapped onto a newly existing surface area of bedrock <6000BP, where possible microbial mat development could have aided in incoming sediment entrapment and further cementing lithification, allowing sediment accumulation to succeed sea level rise in the early Holocene.

5. Island growth occurs as per Figure 7.

#### [...]

#### Works Cited

- AZoM.com. (2016). What is the Effect of Sample Size on Particle Size Distribution Measurements? [online] Available at: https://www.azom. com/article.aspx?ArticleID=12587 [Accessed 28 Oct. 2020].
- Bernier, JC., "Data Dictionary for Grain-Size Data Tables" [online] available at: https://pubs.usgs.gov/ds/0834/data/Metadata/Tables\_ metadata/Cat\_Island\_grainsize\_data%20dictionary.pdf
- Broecker, Wallace S., et al. "Factors Controlling the Rate of CaCO3precipitation on Great Bahama Bank." *Global Biogeochemical Cycles*, vol. 15, no. 3, 2001, pp. 589–596., doi:10.1029/2000gb001350.
- de Mahiques M.M. (2016) Sediment Sorting. In: Kennish M.J. (eds) Encyclopedia of Estuaries. Encyclopedia of Earth Sciences Series. Springer, Dordrecht. https://doi.org/10.1007/978-94-017-8801-4\_348
- data.world. (n.d.). *Cat Island marine core grain-size data from field ... oBIM06 dataset by us-doi-gov*. [online] Available at: https://data.world/us-doi-gov/67bc3fdc-0aed-4054-b75a-2d11e475819c [Accessed 28 Oct. 2020].
- Diaz, Mara & Eberli, Gregor & Blackwelder, Patricia & Phillips, Brian & Swart, Peter. (2017). Microbially mediated organomineralization in the formation of ooids. Geology. 45. 771-774. 10.1130/G39159.1.
- Diaz, Mara R., and Gregor P. Eberli. "Decoding the Mechanism of Formation in Marine Ooids: A Review." *Earth-Science Reviews*, vol. 190, 2019, pp. 536–556., doi:10.1016/j.earscirev.2018.12.016.
- Dickson, J.A.D., and Arthur Saller. "Caicos Platform Shoals Formation and Erosion, Turks and Caicos Islands." *Developing Models and Analogs for Isolated Carbonate Platforms—Holocene and Pleistocene Carbonates of Caicos Platform, British West Indies*, 2008, doi:10.2110/pec.08.22.0133.
- Dravis, Jeffrey J., and Harold R. Wanless. "Impact of Strong Easterly Trade Winds on Carbonate Petroleum Exploration Relationships Developed from Caicos Platform, Southeastern Bahamas." *Marine and Petroleum Geology*, vol. 85, 15 Apr. 2017, pp. 272–300., doi:10.1016/j.marpetgeo.2017.04.010.
- Dravis, Jeffrey J., and Harold R. Wanless, 2008. "Caicos Platform Models of Quaternary Carbonate Deposition Controlled by Stronger Easterly Trade Winds–Applications to Petroleum Exploration", Developing Models and Analogs for Isolated Carbonate Platforms– Holocene and Pleistocene Carbonates of Caicos Platform, British West Indies, William A. Morgan, Paul M. (Mitch) Harris
- Dravis, Jeffrey J., and Harold R. Wanless, 1989. "Carbonate Environments and Sequences of Caicos Platform: Caicos, British West Indies to Miami, Florida", July 20 26, 1989, Volume T<sub>374</sub>
- Duguid, Sarah Marie Anne. "Origin and geochemistry of modern Bahamian ooids." Queens University Masters Thesis, 2009, MR48176
- Dutch, Rian & Krapf, Carmen & Curtis, Stacey & Nicolson, Benjamin & Pawley, Mark & Preiss, Wolfgang & Sheard, Malcolm & Werner, Mario. (2015). Geological Survey of South Australia glossary of Validated lithology modifier and landform terms used for digital field note capture.
- Entegris. 2020. What is the Effect of Sample Size on Particle Size Distribution Measurements?. AZoM, viewed 27 October 2020, https://www.azom.com/article.aspx?ArticleID=12587.
- Jamison-Todd, S., Stein, N., Overeem, I., Khalid, A., & Trower, E. J. (2020). Hurricane deposits on carbonate platforms: A case study of Hurricane Irma deposits on Little Ambergris Cay, Turks and Caicos Islands. *Journal of Geophysical Research: Earth Surface*, 125, e2020JF005597. https://doi.org/10.1029/2020JF005597
- Jones, Gareth D., et al. "An Investigation of Early Diagenesis in an Isolated Carbonate Platform Caicos Platform, British West Indies." Developing Models and Analogs for Isolated Carbonate Platforms—Holocene and Pleistocene Carbonates of Caicos Platform, British West Indies, 2008, doi:10.2110/pec.08.22.0177.

- Kerans, Charles, et al. "Anatomy of a Late Quaternary Carbonate Island: Constraints on Timing and Magnitude of Sea-Level Fluctuations, West Caicos, Turks and Caicos Islands, BWI." *Quaternary Science Reviews*, vol. 205, Jan. 2019, pp. 193–223., doi:10.1016/j.quascirev.2018.12.010.
- Kaczmarek, Stephen & Hasiuk, Franciszek. (2008). Mapping surficial sediment distributions on Caicos Platform: A quantitative approach integrating statistical analysis of Landsat spectral data and field observations. 10.2110/pec.08.22.0057.
- Kindler, Pascal, et al. "Pre-Holocene Island Geology of the Caicos and Mayaguana (Bahamas) Platforms Similarities and Differences." Developing Models and Analogs for Isolated Carbonate Platforms—Holocene and Pleistocene Carbonates of Caicos Platform, British West Indies, 2008, doi:10.2110/pec.08.22.0211.

Krumbein, W. C., 1940, Flood gravels of San Gabriel Canyon, California: Geological Society of America Bulletin, 51:639-676.

- Neese, Douglas G., "OOID DIAGENESIS: HOLOCENE GEOCHEMISTRY OF THE ROCK/WATER/ATMOSPHERIC SYSTEM" University of Oklahoma Ph.D. Dissertation, 1986, 8611788
- Orzechowski, E., Strauss, J., Knoll, A., Fischer, W., Cantine, M., Metcalfe, K., Quinn, D., Stein, N., Gomes, M., and Grotzinger, H., 2016, Age and construction of Little Ambergris Cay bedrock rim, southeastern Caicos Platform, British West Indies, in Proceedings American Geophysical Union Fall Meeting, San Francisco, California, 2016.
- Ostling, Johanna L., et al. "The Biogeomorphology of Mangroves and Their Role in Natural Hazards Mitigation." *Geography Compass*, vol. 3, no. 5, 2009, pp. 1607–1624., doi:10.1111/j.1749-8198.2009.00265.x.
- Powers, M.C., 1953, A new roundness scale for sedimentary particles: Journal of Sedimentary Petrology, 23:117-119.
- Rankey, Eugene C., et al. "Geomorphology and Sedimentology of Ambergris Ooid Shoal, Caicos Platform." *Developing Models and Analogs for Isolated Carbonate Platforms—Holocene and Pleistocene Carbonates of Caicos Platform, British West Indies*, 2008, doi:10.2110/pec.08.22.0127.
- Schlager, Wolfgang. (2003). Benthic carbonate factories of the Phanerozoic. International Journal of Earth Sciences. 92. 10.1007/s00531-003-0327-x.
- Swart, P.k., et al. "The Fertilization of the Bahamas by Saharan Dust: A Trigger for Carbonate Precipitation?" *Geology*, vol. 42, no. 8, 2014, pp. 671–674., doi:10.1130/g35744.1.
- Toscano, M.A. & Macintyre, I.G. 2003, "Corrected western Atlantic sea-level curve for the last 11,000 years based on calibrated 14C dates from Acropora palmata framework and intertidal mangrove peat", Coral Reefs, vol. 22, no. 3, pp. 257-270.
- Trower, Elizabeth J., et al. "Active Ooid Growth Driven By Sediment Transport in a High-Energy Shoal, Little Ambergris Cay, Turks and Caicos Islands." *Journal of Sedimentary Research*, vol. 88, no. 9, 2018, pp. 1132–1151., doi:10.2110/jsr.2018.59.
- Westphal, Hildegard, et al. "Carbonate Depositional Systems: Assessing Dimensions and Controlling Parameters." *Springer*, 2010, doi:10.1007/978-90-481-9364-6.